\documentclass[12pt]{article} 
\usepackage{epsf}
\pagestyle{empty}
\renewcommand{\today}{March 4, 1997}

\special{src: 6 LAB3C.TEX} %Inserted by TeXtelmExtel

% \ilf is inline formula in displaystyle
\newcommand{\ilf}[1]{$\displaystyle #1 $}

\special{src: 11 LAB3C.TEX} %Inserted by TeXtelmExtel


\special{src: 14 LAB3C.TEX} %Inserted by TeXtelmExtel

\begin{document}
\noindent{\bf Math 3226
\hfill
Laboratory  3C}\\
\begin{center}
{\bf Damping and the Harmonic Oscillator}
\bigskip\end{center}

\special{src: 24 LAB3C.TEX} %Inserted by TeXtelmExtel

\noindent Work {\bf Laboratory 2.1} on pp.\ 207--208 of our textbook. 
Use the data set``Choice~3".

\special{src: 28 LAB3C.TEX} %Inserted by TeXtelmExtel

\vspace*{\fill}

\special{src: 32 LAB3C.TEX} %Inserted by TeXtelmExtel

\flushright\tiny\copyright\  H. Knaust. \today .
\end{document}
